Author’s response to reviews

Title: Using Total Quality Management Approach to Improve Patient Safety by Preventing Medication Error Incidences

Authors:

Farah Yousef (farahyousef90@yahoo.com)

Nadin Yousef (ph.nadinyousef@gmail.com)

Version: 2 Date: 20 Jul 2017

Author’s response to reviews:

Dear Editor,

We hope the manuscript to be published in your journal, as we did our best especially for ESL issue. I asked for help from Arabic-English sword translator as my country is in crisis that made most reputable ESL services forbidden for my region.

We have read the Editorial revisions required and they are all DONE as follows:

Comment 1: Please change the title of the Abstract "Objectives" to "Background".

Response 1: Done

Comment 2: In the Declarations:

a) Please change the title of "Ethics Approval and Consent" to Ethics approval and consent to participate

Response 2-a: Done

b) Please change the title of "Publication consent" to Consent to publish. Please also remove "in this section" from the text.

Response 2-b: Done
c) In the Availability of data and materials section, please clarify where the data supporting your findings can be found (including information found in any supplementary files). If you do not wish to share your data, please state (in the ‘Availability of data and Materials’ section) that data will not be shared, and state the reason. If all the data supporting your findings is contained within the manuscript, please state so in the “Availability of data and materials” section. Please note that identifying/confidential patient data should not be shared. Please see http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials for further guidelines.

Response 2-c: They are contained in the text.

Comment 3: Please remove the figure legends from within the figure files, these should contain the image graphic only (and be separate documents for each Figure). Please place the caption as standard text after the references.

Response 3: Done

Comment 4: Please remove your response to reviewers before resubmitting your manuscript.

Response 4: I will do so.

Reviewer 1 (Terri Warholak) comment: Thank you for addressing my comments. However, the English still needs MUCH editing and many sections of the paper are therefore unclear.

Response to Terri Warholak (Reviewer 1)

Comments:

Thank you for your comments. For English revision, I have asked the help of Arabic-English sword translator. This was because reputable ESL services are forbidden in my region due to the crisis my country is going through. Hope you be satisfied about the final revised manuscript.

Thank you very much,

Farah Yousef